Author's response to reviews

Title: Procalcitonin (PCT) and C-reactive Protein (CRP) as severe systemic infection markers in febrile neutropenic adults.

Authors:

Karin S.R. Massaro (rukarin.massaro@uol.com.br)
Silvia F Costa (costasilviaf@ig.com.br)
Claudio Leone (sccihsecretaria@hcnet.usp.br)
Dalton AF Chamone (sccihsecretaria@hcnet.usp.br)

Version: 2 Date: 16 August 2007

Author's response to reviews:

You provided a covering letter for the most recent version of this manuscript on 14 April 2007:

Sao Paulo, 14th April 2007

To

Editor-in-Chief,

We would like to submit the manuscript entitled "PROCALCITONIN (PCT) AND C-REACTIVE PROTEIN (CRP) AS INDICATORS OF INFECTION IN FEBRILE NEUTROGENIC ADULTS" to the section original articles. This manuscript was not submit to elsewhere and all the authors agreed with this submission.

Best Regards,

Silvia F. Costa

MD, PhD, University of Sao Paulo, Brazil.

e-mail: costasilviaf@ig.com.br